Author's response to reviews

Title: Takayasu's arteritis associated with Crohn's disease: report of a case

Authors:

Nasser E Daryani (nebrahim@sina.tums.ac.ir)
Atoosa Nayer Habibi (anayery70@yahoo.com)
Mohammad Bashashati (bashashati_md@yahoo.com)
Moezedin Javad Rafiee (ajdarkosh1345@yahoo.com)
Hossein Ajdarkosh (ajdarkosh1345@yahoo.com)
Mehrdad Azmi (mehrdad_azmi@hotmail.com)

Version: 3 Date: 27 June 2007

Author's response to reviews: see over
Dear Editor in chief of Journal of Medical Case Reports,

Hereby, I would like to submit a manuscript entitled "Takayasu's arteritis associated with Crohn's disease: report of a case" in your scientific journal. In Iran and eastern countries Inflammatory Bowel Disease (IBD) is less common than western area, and by increasing in number of IBD cases in these regions we have to expect to encounter with more extra intestinal comorbidities. And to our knowledge this is the 1st case from Iran.

We have not submitted this manuscript elsewhere and we do not consider sending it to another journal until receiving your response. My co-authors and I have seen and approved this version of the manuscript. Also, we tried to review all same published articles. I hope that it could be published in your journal.

Best regards,
Atoosa Nayerhabibi
Researcher of Gastroenterology, Imam Khomeini Hospital,
Tehran University of Medical Sciences, Iran
EMAIL: anayery70@yahoo.com